Author’s response to reviews

Title: First line in psychiatric emergency: Pre-hospital emergency protocol for mental disorders in Iran

Authors:
Fatemeh Shirzad (Shirzad.f@iums.ac.ir)
Fateme Hadi (Hadi.f@iums.ac.ir)
Seyede Salehe Mortazavi (salehe.mortazavi@gmail.com)
Maryam Biglari (dr.maryam54@gmail.com)
Hassan Noori Sari (hassannourisari@yahoo.com)
Zeinab Mohammadi (m.mohammady35@gmail.com)
Mehrdad Kazemzade Atoofi (atoofi.m@iums.ac.ir)
Seyed Vahid Shariat (vahidshariat@gmail.com)

Version: 3 Date: 21 Feb 2020

Author’s response to reviews:

Dear Editor,

The manuscript was revised based on your comments:

1-The abbreviations were used in the text defined in the text at first use, and a list of abbreviations provided in the Declarations.
2-In the section ‘Funding’, the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript was described.
3- The part response to the reviewers was removed.
4- Manuscript as a single, final, clean version was uploaded.

Other comments were done before.

Best wishes,

Mortazavi